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WEATHER COMMENT: Fri May 12 16:26:59 GMI 1995

Col d, in heavy altocunulus overcast, clouds broken, but clear holes
of significant size very sparse, strong southeast w nd approx. 30 nph
continues. Tenp=46F.

COWMMENT: Fri May 12 16:28:32 GMI 1995
Continuing effort to repair spar guider.

COMMENT: Fri May 12 21:57:44 GV 1995

Found the problem after revisiting the notor. The gears are stripped
on both the on that was in use and in the spare--sone spare huh?
Thanks to Howard Hull’s gui dance, the troubl eshooting went a | ot
faster by pronpting me to question sone of my test results from
yesterday and taking a second | ook at many of the tests | ran.

Howard will look thru their spares today for a good spare and

try calling Vernitron ASAP (they’'re closed already) for a new one

if one can’t be located at HAO

COMMENT: Fri May 12 22:05:12 GMI 1995

Put everything back into place, except for notor and access covers.

No observations. Another green dot day.

No observations tonorrow. DAK on vacation and EAY will regroup.

WIIl check email and LOA operation on Sunday. Doubt part will be here
by then.

WEATHER COMMVENT: Fri May 12 22:08:47 GMI 1995
Still in clouds--not mssing any obs anyway.

COMMENT: Fri May 12 22:09: 08 GMI 1995
Tapes: LOAL: L00235

Fri May 12 22:09:48 GMI 1995: Fil emark



